Author's response to reviews

Title: Stage I seminoma: Treatment outcomes at King Hussein Cancer Center in Jordan

Authors:

Jamal Khader (jkhader@khcc.jo)
Ahmed Salem (alwikah@hotmail.com)
Yazan Abuodeh (yabuodeh@khcc.jo)
Abdelatief Almousa (aalmousa@khcc.jo)
Naim Farah (nfarah@khcc.jo)
Fadwa Abdelrahman (fadelrahman@khcc.jo)

Version: 5 Date: 11 April 2012

Author's response to reviews: see over
Dear Dr. Rosser,

Thank you for reviewing our submitted manuscript entitled “Stage I seminoma: Treatment outcomes at King Hussein Cancer Center in Jordan”.

Below are the responses to the reviewer’s comments. Also find the enclosed revised manuscript. All revisions were made in red font.

Thank you for considering this manuscript for publication.

Kind regards,

Jamal Khader MD

King Hussein Cancer Center

HM Queen Rania Street

P.O.Box: 1269, Amman 11941

Amman, Jordan

Tel.: +962 788 305344

Email: jkhader@khcc.jo
Reviewer: Ahmed Mansour

Comments: None

Reviewer: Kazunori Namiki

Minor Essential Revisions
It is better to change the title to "Stage I seminoma: Treatment outcomes at King Hussein Cancer Center in Jordan"
The title has been changed

Quality of written English: Needs some language corrections before being published
The manuscript was revised and necessary minor corrections were made